%% if you are submitting an initial manuscript then you should have submission as an option here
%% if you are submitting a revised manuscript then you should have revision as an option here
%% otherwise options taken by the article class will be accepted
\documentclass[submission]{FPSAC2019}
%% but DO NOT pass any options (or change anything else anywhere) which alters page size / layout / font size etc

\articlenumber{20}
\addbibresource{20.bib}

%% note that the class file already loads {amsmath, amsthm, amssymb}

\usepackage{graphics,array}
\usepackage{tikz}
\usetikzlibrary{matrix,arrows,patterns,cd,shapes}
\usepackage{mathtools}
\usepackage{amsfonts}

 \newtheorem{theorem}{Theorem}[section]
 \newtheorem{corollary}[theorem]{Corollary}
 \newtheorem{prop}[theorem]{Proposition}
 \newtheorem{conjecture}[theorem]{Conjecture}
 \newtheorem{lemma}[theorem]{Lemma}
 \theoremstyle{definition}
 \newtheorem{example}[theorem]{Example}
 \newtheorem{observation}[theorem]{Observation}
 \newtheorem{question}[theorem]{Question}
 \newtheorem{problem}[theorem]{Problem}
 \newtheorem{definition}[theorem]{Definition}
 \newtheorem{remark}[theorem]{Remark}
 \newtheorem{characterization}[theorem]{Characterization}
 \theoremstyle{remark}
 \newtheorem{rem}[theorem]{Remark}


\newcommand{\GETOUT}[1]{}
 
 %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% BEGIN our macros
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\newcommand{\flip}[1]{\mathsf{flip}(#1)}
\newcommand{\asc}{\mathsf{asc}}
\newcommand{\dire}[2]{{#1}_{(#2)}}

\newcommand{\bij}{f}
\newcommand{\bam}{\bij_{\mathsf{AM}}}  % (b)ijection (a)scent_sequences to (m)atrices
\newcommand{\bma}{\bij_{\mathsf{MA}}}
\newcommand{\bap}{\bij_{\mathsf{AP}}}
\newcommand{\bas}{\bij_{\mathsf{AS}}} % S for {s}ymmetric group
\newcommand{\bpa}{\bij_{\mathsf{PA}}}
\newcommand{\bmp}{\bij_{\mathsf{MP}}}
\newcommand{\bpm}{\bij_{\mathsf{PM}}}
\newcommand{\bps}{\bij_{\mathsf{PS}}}

\newcommand{\am}[1]{$\mathsf{AM#1}$}
\newcommand{\ma}[1]{$\mathsf{MA#1}$}
\newcommand{\ap}[1]{$\mathsf{AP#1}$}
\newcommand{\pa}[1]{$\mathsf{PA#1}$}

\newcommand{\Asc}{\mathsf{Asc}}
\newcommand{\RAsc}{\mathsf{RAsc}}
\newcommand{\CAsc}{\mathsf{CAsc}}
\newcommand{\Posets}{\mathsf{Posets}}
\newcommand{\RPosets}{\mathsf{RPosets}}
\newcommand{\CPosets}{\mathsf{CPosets}}
\newcommand{\Matrices}{\mathsf{Matrices}}
\newcommand{\RMatrices}{\mathsf{RMatrices}}
\newcommand{\CMatrices}{\mathsf{CMatrices}}
\newcommand{\Perms}{\mathsf{Perms}} % Other option: $\sn(2|3\overline{1})$
\newcommand{\RPerms}{\mathsf{RPerms}} % Other option: $\sn(3\overline{1}52\overline{4})$
\newcommand{\CPerms}{\mathsf{CPerms}}  %Other option:  $\sn(231)$
\newcommand{\sn}{\mathcal{S}_n}
\newcommand{\RGF}{\mathsf{RGF}}
\newcommand{\Structure}{\mathsf{Struct}}
\newcommand{\reverse}{\mathrm{rev}}
\renewcommand{\complement}{\mathrm{comp}}
\newcommand{\D}{\mathsf{pan}}
\newcommand{\R}{\mathsf{R}}
\newcommand{\C}{\mathsf{C}}

\newcommand{\level}{\ell}
\newcommand{\eell}{{l}}
\newcommand{\minlevel}{\mathrm{\ell}^{\star}}
\newcommand{\remove}{\mathsf{remove}}
\newcommand{\addition}{\mathsf{add}}
\newcommand{\mindex}{\mathsf{mindex}}
\newcommand{\reduce}{\mathsf{reduce}}
\newcommand{\rowsum}{\mathsf{rowsum}}
\renewcommand{\max}{\mathsf{max}}

\newcommand{\A}{\mathbf{a}}
\renewcommand{\a}{a}

 %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Numbering
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%\numberwithin{equation}{section}
%\numberwithin{figure}{section}
\renewcommand{\labelitemi}{$\circ$} 
\renewcommand{\theenumi}{\upshape{(}\alph{enumi}\upshape{)}}
\renewcommand{\labelenumi}{\theenumi}



%\subjclass[2010]{05A19 (Primary); 05A05, 06A06 (Secondary)} 





%% define your title in the usual way
\title[Bijections for ascent sequences]{Refining the bijections among ascent sequences, (2+2)-free posets, integer matrices and pattern-avoiding permutations}

%% define your authors in the usual way
%% use \addressmark{1}, \addressmark{2} etc for the institutions, and use \thanks{} for contact details
\author{Mark Dukes\thanks{\href{mailto:mark.dukes@ucd.ie}{mark.dukes@ucd.ie}. Mark Dukes was supported by EPSRC grant EP/M015874/1.}\addressmark{1} \and Peter R.\,W. McNamara\thanks{\href{mailto:peter.mcnamara@bucknell.edu}{peter.mcnamara@bucknell.edu}. Peter McNamara was supported by Simons Foundation grant \#245597.}\addressmark{2}}

%% then use \addressmark to match authors to institutions here
\address{\addressmark{1}UCD School of Mathematics and Statistics, University College Dublin, Dublin 4, Ireland\\ \addressmark{2}Department of Mathematics, Bucknell University, Lewisburg, PA 17837, USA}

%% put the date of submission here
\received{\today}

%% leave this blank until submitting a revised version
%\revised{}

%% put your English abstract here, or comment this out if you don't have one yet
%% please don't use custom commands in your abstract / resume, as these will be displayed online
%% likewise for citations -- please don't use \cite, and instead write out your citation as something like (author year)
\abstract{The combined work of Bousquet-M\'elou, Claesson, Dukes, Jel\'inek, Kitaev, Kubitzke and Parviainen has resulted in non-trivial bijections among ascent sequences, (2+2)-free posets, upper-triangular integer matrices, and pattern-avoiding permutations.  To probe the finer behavior of these bijections, we study two types of restrictions on ascent sequences.  These restrictions are motivated by our results that their images under the bijections are natural and combinatorially significant.  In addition, for one restriction, we are able to determine the effect of poset duality on the corresponding ascent sequences, matrices and permutations, thereby answering a question of the first author and Parviainen in this case.  The second restriction should appeal to \linebreak
Catalaniacs.
}

%% put your French abstract here, or comment this out if you don't have one
%\resume{\lipsum[2]}

%% put your keywords here, or comment this out if you don't have them yet
\keywords{ascent sequence, (2+2)-free poset, interval order, upper-diagonal matrix, pattern avoidance, series-parallel poset}

%% you can include your bibliography however you want, but using an external .bib file is STRONGLY RECOMMENDED and will make the editor's life much easier
%% regardless of how you do it, please use numerical citations, ie. [xx, yy] in the text

%% this sample uses biblatex, which (among other things) takes care of URLs in a more flexible way than bibtex
%% but you can use bibtex if you want
%\usepackage[backend=bibtex]{biblatex}
%\addbibresource{sample.bib}
%% note the \printbibliography command at the end of the file which goes with these biblatex commands

%\addbibresource{biblio.bib}

\begin{document}
\begin{thebibliography}{99}

%\bibitem{AndJel} G.E. Andrews and V. Jel\'inek. On $q$-Series identities related to interval orders. {\it{Eur. J. Combin.}} {\bf{39}} (2014), 178--187.

% \bibitem{Bog} K.P. Bogart.  An obvious proof of Fishburn's interval order theorem.  {\it{Discrete Math.}} {\bf{118}} (1993), no.1--3, 239--242.

%\bibitem{bcdk} M. Bousquet-M\'elou, A. Claesson, M. Dukes and S. Kitaev. (${\mathbf{2+2}}$)-free posets, ascent sequences and pattern avoiding permutations. {\it{J. Comb. Theory Ser. A}} {\bf{117}} (2010), no.~7, 884--909.

%\bibitem{cyz} D. Chen, S.H.F. Yan, R.D.P. Zhou. Equidistributed statistics on Fishburn matrices and permutations.  Preprint.  \href{https://arxiv.org/abs/1808.04191}{\texttt{arXiv:1808.04191}}.

%\bibitem{cdkub} A. Claesson, M. Dukes and M. Kubitzke. Partition and composition matrices. {\it{J. Comb. Theory Ser. A}}, {\bf{118}} (2011), no.~5, 1624--1637.

%\bibitem{dfpr10} F. Disanto, L. Ferrari, R. Pinzani and S. Rinaldi.  Catalan pairs: A relational-theoretic approach to Catalan numbers. {\it{Adv. App. Math.}} {\bf{45}} (2010), 505--517.

%\bibitem{dkrs} M. Dukes, S. Kitaev, J. Remmel and E. Steingr\'imsson.  Enumerating (2+2)-free posets by indistinguishable elements. {\it{Journal of Combinatorics}} {\bf{2}} (2011), no.~1, 139--163.

%\bibitem{djk} M. Dukes, V. Jel\'inek and M. Kubitzke.  Composition matrices, (2+2)-free posets and their specializations.  {\it{Electron. J. Comb.}} {\bf{18}} (2011), no.~1,  P44, 9 pp.

%\bibitem{dukmcn} M. Dukes and P.R.W. McNamara. Refining the bijections among ascent sequences, (2+2)-free posets, integer matrices and pattern-avoiding permutations. Preprint.  \href{https://arxiv.org/abs/1807.11505}{\texttt{arXiv:1807.11505}}.

%\bibitem{dp} M. Dukes and R. Parviainen. Ascent sequences and upper triangular matrices containing non-negative integers. {\it{Electron. J. Comb.}} {\bf{17}} (2010), no.~1,  R53, 16 pp.

%\bibitem{ds} P. Duncan and E. Steingr\'imsson.  Pattern avoidance in ascent sequences.  {\it{Electron. J. Comb.}} {\bf{18}} (2011), no.~1, Paper 226, 17 pp.

%\bibitem{Gor} G. Gordon.  Series-parallel posets and the Tutte polynomial.  {\it{Discrete Math.}} {\bf{158}} (1996), no.~1--3, 63--75.

%\bibitem{Gua} M. Guay-Paquet. A modular relation for the chromatic symmetric functions of (3+1)-free posets. Preprint. \href{https://arxiv.org/abs/1306.2400}{\texttt{arXiv:1306.2400}}.

%\bibitem{JelSelfdual} V. Jel\'inek. Counting general and self-dual interval orders. {\it{J. Comb. Theory Ser. A}} {\bf{119}} (2012), no.~3, 599--614.

%\bibitem{JelFishburn} V. Jel\'inek. Catalan pairs and Fishburn triples. {\it{Adv. App. Math.}} {\bf{70}} (2015), 1--31.

%\bibitem{ky} M.T. Keller and S.J. Young.  Hereditary semiorders and enumeration of semiorders by dimension.  Preprint.  \href{https://arxiv.org/abs/1801.00501}{\texttt{arXiv:1801.00501}}.
%
%\end{thebibliography}


\maketitle
%% note that you DO NOT have to put your abstract here -- it is generated by \maketitle and the \abstract and \resume commands above


\section{Introduction}\label{sec:intro}

In the last decade, an interesting collection of results has emerged from the study of (2+2)-free posets, or interval orders as they are also known,
and their connection to permutations avoiding a non-standard permutation pattern of length three.
The starting point for this story was the introduction in Bousquet-M\'elou et al.~\cite{bcdk} of a new type of permutation pattern that the authors termed a {\it{bivincular pattern}}.
In that paper it was proven that length-$n$ permutations avoiding the bivincular pattern
$2|3\overline{1}$ were in one-to-one correspondence with unlabelled (2+2)-free posets on $n$ elements.
This was shown by encoding both structures as an integer sequence of length $n$ that has come to be known as an {\it{ascent sequence}}.  Via ascent sequences, Bousquet-M\'elou et al.\ were able to solve the long-standing open problem of enumerating unlabelled (2+2)-free posets.

In \cite{dp}, these ascent sequences were shown to uniquely encode another set of objects:
all square upper-triangular matrices of non-negative integers whose entries sum to $n$ which have neither rows nor columns consisting of only zeros.

These initial two papers linking (bijectively) four different discrete objects led to a series of papers that studied these bijections and built upon the correspondences.
From the enumerative viewpoint, Dukes et al.~\cite{dkrs} 
considered these objects according to several statistics and presented multivariate generating functions for these statistics.  
Two additional papers \cite{cdkub,djk} have studied analogues of these bijections on \emph{labeled} posets.

In another direction, and more recently, it has emerged that refinements of these correspondences have equally compelling stories to tell.
Duncan \& Steingr\'imsson~\cite{ds} studied pattern avoidance in ascent sequences and established bijections between pattern avoiding ascent sequences and other combinatorial objects such as set partitions and objects enumerated by the Catalan and Narayana numbers.
Jel\'inek~\cite{JelSelfdual} presented a new method to derive formulas for the generating functions of interval orders.
The method generalised the results of~\cite{dkrs}
and also allowed the enumeration of self-dual interval orders with respect to several statistics.
Using his newly derived generating function formulas, Jel\'inek proved a bijective relationship between self-dual interval orders and upper-triangular matrices having no zero rows~\cite{JelSelfdual}.  
Andrews \& Jel\'inek~\cite{AndJel} built on Jel\'inek's work and proved several power series identities involving the refined generating functions for interval orders and self-dual interval orders.
Keller and Young~\cite{ky} considered the difficult question of determining which ascent sequences map to semiorders; also known as unit interval orders, semiorders are posets that are both (2+2) and (3+1)-free. 
Most recently, \cite{cyz} uses a slightly different bijection to resolve conjectures of Jel\'inek about the equistribution of certain statistics in $2|3\overline{1}$-avoiding permutations and the class of matrices described above.  

This extended abstract summarizes results in our recent paper~\cite{dukmcn}. It adds to this body of work by analyzing two types of restrictions on ascent sequences.
One motivation for these restrictions is that their images through the bijections of~\cite{bcdk,djk,dp} are combinatorially significant in the subsets they identify,
e.g. series-parallel posets and 231-avoiding permutations.
Moreover, the analysis of the images of these ascent sequences allows us to prove results about duals of each of the structures,
thus going some way in answering an open problem of Dukes \& Parviainen~\cite{dp}.  In our restricted settings, the bijections of~\cite{bcdk,djk,dp} coincide with those of~\cite{cyz}.

The first type of restriction we study (in \cref{sec:restricted_sets}) begins with a restriction on the types of ascents one may have in an ascent sequence.  
In particular, when the bijection of~\cite{bcdk} recursively builds a (2+2)-free poset from an ascent sequence, there are some ascents that cause complicated and unnatural modifications to the poset, while the bijection treats all other ascents in a very natural way.  Our first restriction is to those ascent sequences that contain only these ascents that result in this latter natural behaviour.  A motivation for this restriction is that this good behaviour carries through to the general framework of bijections.  Indeed, the images of these new \emph{restricted ascent sequences} $\RAsc$ through the bijections given in~\cite{bcdk,djk,dp} are proven to be simple restrictions:
the subset $\RMatrices$ of the matrices from~\cite{dp} having only positive diagonal entries,
the subset $\RPosets$ of (2+2)-free posets which have a chain 
of the maximal possible length, and the set $\RPerms$ of permutations avoiding the barred pattern $3 \overline{1}5 2\overline{4}$. This set $\RPerms$ was already identified in~\cite{bcdk} in the context of modified ascent sequences.  See \cref{fig:summary} for a diagram outlining our sets and maps of interest.
 
In \cref{sec:duality} we give a partial solution to an open problem of Dukes \& Parviainen~\cite{dp} by addressing the topic of structural duality.
The dual $P^*$ of a (2+2)-free poset $P$ is also a (2+2)-free poset.
This observation prompts the question as to whether one can derive the (ascent sequence/matrix/permutation) corresponding to $P^*$ from the (ascent sequence/matrix/permutation)  corresponding to $P$.  In the matrix case, \cite{djk} provides the answer; see \cref{exa:matrix_dual} below.  In the ascent sequence and permutation cases, this question seems intractable in general because of the complicated map between some ascent sequences and posets as mentioned in the previous paragraph.  However, consistent with our motivation for restricting to better behaved sets, we can answer this duality question completely for all posets in $\mathsf{RPosets}$, which we do in \cref{sec:duality}.

In \cref{sec:catalan}, we consider the Catalan family $\CAsc$ of 101-avoiding ascent sequences studied in~\cite{ds}, and investigate their images under the bijections of~\cite{bcdk,djk,dp}.  
The results are perhaps even nicer than the $\R$-families and are shown in \cref{fig:summary}.  The posets that arise are the series-parallel interval orders, i.e., those that are both (2+2)-free and N-free.  This class of posets appears in \cite{dfpr10},
while series-parallel posets in general are widespread in the literature, partially because their recursive structure permits many polynomial-time algorithms (see, for example, \cite{Gor} and the references therein).  The matrices and permutations which correspond to these ascent sequences are those matrices from $\RMatrices$ that are termed \emph{SE-free} in \cite{JelFishburn}, and 231-avoiding permutations, respectively.


\begin{figure}
\begin{center}
\footnotesize
%\begin{tikzpicture}[scale=0.5, rounded corners = 5pt]
\begin{tikzpicture}[scale=0.51, rounded corners = 5pt]
\begin{scope}
%\draw (0,7.5) node {$\Posets$};
\draw (0,6.05) node {$\Posets$:\,(2+2)-free posets $P$};
\draw (0,4.5) node {$\RPosets$: $P$ has a};
\draw (0,3.6) node {chain of length $\level(P)$};
\draw (0,2.2) node {$\CPosets$:};
\draw (0,1.3) node {series-parallel $P$};
\draw (-5,0) rectangle (5,7);
\draw (-4,0.2) rectangle (4,5);
\draw (-3,0.4) rectangle (3,3);
\end{scope}
\begin{scope}[xshift = 82ex]
%\draw (0,7.5) node {$\Asc$};
\draw (0,6.05) node {$\Asc$: ascent sequences $\A$};
\draw (0,4.05) node {$\RAsc$: self-modified $\A$};
\draw (0,2.2) node {$\CAsc$:};
\draw (0,1.3) node {$101$-avoiding $\A$};
\draw (-5,0) rectangle (5,7);
\draw (-4,0.2) rectangle (4,5);
\draw (-3,0.4) rectangle (3,3);
\end{scope}
\begin{scope}[yshift = -66ex]
%\draw (0,7.5) node {$\Matrices$};
\draw (0,6.5) node {$\Matrices$: upper-$\triangle$ $\mathbb{N}$-matrices};
\draw (0,5.6) node {$M$ w/o empty rows or columns};
\draw (0,4.5) node {$\RMatrices$: $M$ has only};
\draw (0,3.6) node {positive diagonal entries};
\draw (0,2.2) node {$\CMatrices$:};
\draw (0,1.3) node {$M$ is SE-free};
\draw (-5,0) rectangle (5,7);
\draw (-4,0.2) rectangle (4,5);
\draw (-3,0.4) rectangle (3,3);
\end{scope}
\begin{scope}[xshift = 82ex, yshift = -66ex]
%\draw (0,7.5) node {$\Perms$};
\draw (0,6.05) node {$\Perms$: $\sn(2|3\overline{1})$};
\draw (0,4.05) node {$\RPerms$: $\sn(3\overline{1}52\overline{4})$};
\draw (0,1.7) node {$\CPerms$: $\sn(231)$};
\draw (-5,0) rectangle (5,7);
\draw (-4,0.2) rectangle (4,5);
\draw (-3,0.4) rectangle (3,3);
\end{scope}
\draw[->] (-1,-1) -- (-1,-3);
\draw (-1.7,-2) node {$\bpm$};
\draw[<-] (1,-1) -- (1,-3);
\draw (1.7,-2) node {$\bmp$};
\draw[->] (5.8,4.5) -- (7.8,4.5);
\draw (6.8,5) node {$\bpa$};
\draw[<-] (5.8,2.5) -- (7.8,2.5);
\draw (6.8, 2) node {$\bap$};
\draw[->] (13.7,-1) -- (13.7,-3);
\draw (13.0,-2) node {$\bas$};
\draw[->] (5.2,-2.8) -- (7.2,-0.8);
\draw (5.7, -1.3) node {$\bma$};
\draw[<-] (6.2,-3.7) -- (8.2,-1.7);
\draw (7.75, -3.05) node {$\bam$};
\end{tikzpicture}
\end{center}
\caption{A diagrammatic summary of the sets and bijections of interest.}
\label{fig:summary}
\end{figure}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Preliminaries}\label{sec:preliminaries}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

While the intimate connections between four different types of objects are certainly a strength of this area of study, the drawback for our present purposes is that there is a considerable amount of background that needs to be introduced, including all four classical sets and many of the bijections among them.  Our use of the words ``classical'' refers to the full sets considered by most of the papers mentioned in the Introduction, and as shown by the largest boxes in \cref{fig:summary}:  ascent sequences, (2+2)-free posets, upper-triangular matrices with non-negative integer entries having neither rows nor columns of all zeros, and permutations avoiding $2|3\overline{1}$.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsection{The classical sets}\label{sub:classicalsets}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

An \emph{ascent sequence} is a sequence $\A = (\a_1, \ldots,\a_n)$ of non-negative integers such that $\a_1=0$, and for all $i$ with $1<i \leq n$ we have $\a_i \leq \asc(\a_1,\ldots,\a_{i-1}) +1$, where $\asc(\a_1,\ldots,\a_k)$ denotes the number of ascents in the sequence $(\a_1,\ldots,\a_k)$.  For example $(0,1,0,1,3)$ is an ascent sequence whereas $(0,1,0,2,4)$ is not.  Let $\Asc_n$ be the set of all ascent sequences of length $n$, and let $\Asc$ denote the union of these sets over all $n$, with the same convention applying to all the notation below when the subscript $n$ is dropped.

Let $\Posets_n$ be the set of (2+2)-free posets on $n$ elements, meaning posets that have no induced subposet isomorphic to a disjoint union of two 2-element chains.  We will be interested in a different defining property of (2+2)-free posets, as we now describe.  Let $P=(P,\preceq)$ be a poset with $n$ elements.  
Given $x \in P$, the set $D(x)=\{y \in P : y \prec_P x\}$ is called the \emph{strict downset} of $x$.  
A fact described as ``well-known'' in~\cite{Bog} and which is easy to check is that a poset is (2+2)-free if and only if the set of strict downsets of elements of $P$ can be linearly ordered by inclusion.  
We let $\level(P)$ denote the number of distinct nonempty such downsets, so that $D(P)=(D_0,\ldots,D_{\level(P)})$ is the sequence of downsets of $P$ linearly ordered by inclusion.  In other words $\emptyset = D_0 \subsetneq D_1 \subsetneq \ldots \subsetneq D_{\level(P)}$.  For example, for the poset $P$ in the top left of \cref{fig:examples}, we have
\[
D(P) = (\emptyset, \{p_1,p_2\}, \{p_1, p_2, p_5\}, \{p_1, p_2, p_3, p_5\}).
\]
(Note that while this example $P$ is labelled for the purposes of the explanation, the elements of $\Posets$ are unlabelled.)
We will call $D_i$ \emph{level} $i$ of $P$, and an element $x \in P$ with $D(x)=D_i$ for some $i$ will be said to \emph{lie at level} $i$ of $P$.
Let $L_i=L_i(P)$ be the set of elements lying at level $i$ of $P$ and set $L(P)=(L_0,\ldots,L_{\level(P)})$.
Again for $P$ from \cref{fig:examples}, 
\[
L(P) = (\{p_1, p_2, p_5\}, \{p_6\}, \{p_3\}, \{p_4\}).
\]

Let $\Matrices_n$ be the set of all upper-triangular matrices of indeterminate dimension whose entries are all non-negative integers such that there is neither a row nor a column containing only zeros, and whose sum of all entries is $n$.  Throughout, the entries of the matrix $M$ will be denoted $m_{ij}$ or $m_{i,j}$ but we use $M_{ij}$ when the entries of the matrix are sets.


A sequence $\A = (\a_1, \ldots, \a_r)$ of non-negative integers is said to \emph{contain} a sequence $\mathbf{b} =(b_1, \ldots, b_s)$ \emph{as a pattern} if there exists a subsequence of $\A$ of length $s$ whose elements are in the same relative order as those of $\mathbf{b}$.  We say $\A$ is \emph{$\mathbf{b}$-avoiding} is it does not contain $\mathbf{b}$.  For example, $(0,2,1,3,1,0,2)$, which we write as 0213102 for short, contains the pattern $0101$ because of its subsequence $0202$, but avoids the pattern $1010$.  

When considering pattern-avoidance in sequences that are permutations, we allow for a more general notion of pattern: a permutation $\pi = \pi_1\ldots \pi_n$ is said to \emph{contain} the pattern $2|3\overline{1}$ if there exists an occurrence $\pi_i \pi_j \pi_k$ of 231 in $\pi$ with the additional conditions that $j=i+1$ and $\pi_i = \pi_k+1$.  For example, $32541$ contains $2|3\overline{1}$ because of the occurrence $251$ of $231$, whereas $31452$ avoids $2|3\overline{1}$ even though it has three occurrences of the classical pattern $231$.  The pattern $2|3\overline{1}$ is an example of a \emph{bivincular pattern} as introduced in~\cite{bcdk} since it puts conditions on both the entries and positions of an occurrence.  As usual, we let $\sn(2|3\overline{1})$ denote the set of permutations of length $n$ that avoid $2|3\overline{1}$, and this is exactly our set $\Perms_n$.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsection{The bijections}\label{sub:bijections}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

The results we present concern a variety of combinatorial objects and several bijections between them.
These bijections are the `classical' bijections that were presented in the papers~\cite{bcdk,djk,dp}.
Given the size limitation for this extended abstract, we will present just one of these bijections here as a way to give a sense of their non-triviality.  An exposition of the full set of bijections can be found in the companion paper~\cite[Sec. 2.2]{dukmcn} where it requires five pages.
We will introduce the following convention:
to denote the bijections from~\cite{bcdk,djk,dp}, we will use labels according to their domain and codomain, but rather than use the labels $\Asc$, $\Posets$, $\Matrices$ and $\Perms$, we use the single-letter subscripts $\mathsf{A}$, $\mathsf{P}$, $\mathsf{M}$ and $\mathsf{S}$ (``$\mathsf{S}$'' for ``symmetric group'').   
For example, $\bap$ denotes the bijection of~\cite{bcdk} from ascent sequences to (2+2)-free posets, which is the bijection we choose to present in full.  An example of all the bijections is given in \cref{fig:examples}.

We use the notation $[a,b]$, if $a<b$ are integers, for the set $\{a, \ldots,b\}$ and $[a,b)$ for the set $\{a,\ldots,b-1\}$, etc.  Given a (2+2)-free poset $P$, recall that $\level(P)$ denotes the highest index of a level. % or, equivalently, $\level(P)+1$ is the  number of levels in $P$.  
Let $\minlevel(P)$ denote the minimum index of a level that contains a maximal element.  

\begin{definition}\label{def:bap}
Given $\A = (\a_1, \ldots, \a_n) \in \Asc_n$, we define $\bap(\A)$ recursively.  Let $\A^{(k)}:=(\a_1,\ldots,\a_k)$.
First, $\bap(\A^{(1)})$ 
is the poset consisting of a single element $p_1$.  Supposing $P^{(k)} = \bap(\A^{(k)})$ for some $k \in [1,n)$, we have the following three cases for defining $P^{(k+1)} = \bap(\A^{(k+1)})$.
\begin{enumerate}
\item[\ap1]  If $\a_{k+1} \in [0,\minlevel(P^{(k)})]$ then let $P^{(k+1)}$ be the result of adding to $P^{(k)}$ a new maximal element $p_{k+1}$ that covers 
the same elements as do the elements in 
$L_{a_{k+1}}(P^{(k)})$.
\item[\ap2] If $\a_{k+1}=1+\level(P^{(k)})$ then let $P^{(k+1)}$ be the result of adding to $P^{(k)}$ a new element $p_{k+1}$ covering all maximal elements of $P^{(k)}$. 
\item[\ap3] If $\a_{k+1} \in (\minlevel(P^{(k)}), \level(P^{(k)})]$ 
then let $P^{(k+1)}$ be the outcome of the following: to $P^{(k)}$, 
add a new element $p_{k+1}$ covering the same elements as the elements in $L_{a_{k+1}}(P^{(k)})$.  Let $\mathcal{M}$ be the set of maximal elements of $P^{(k)}$ lying at any level less than $\a_{k+1}$.  
Add all relations $x \preceq y$ where $x \in \mathcal{M}$ and $y$ is any element of $L_{a_{k+1}}(P^{(k)}) \cup \cdots \cup L_{\level(P^{(k)})}(P^{(k)})$; 
here we do not consider the new element $p_{k+1}$ to be an element of $L_{a_{k+1}}(P^{(k)})$.
\end{enumerate}
Then $\bap(\A) = \bap(\A^{(n)}) = P^{(n)}$.  
\end{definition}

\begin{example}\label{exa:bap}
Let $\A = (0,0,1,2,0,1)$ as in \cref{fig:examples}.  Certainly, $P^{(1)}$ is the one-element poset.  The recursive construction of $P = \bap(\A)$ appears in \cref{fig:bapexample}, where the dotted shapes depict the different levels.  The element $a_{k+1}$ appears above each arrow and the case name appears below each arrow. The labels $p_i$ are just for expository purposes and are not part of $\bap(\A)$.  In the final step, $\mathcal{M} = \{p_5\}$.  Note that the new element $p_6$ in the final step ends up on its own level, and this is true in general for applications of \ap3.
\end{example}

\begin{figure}
\begin{center}
\begin{tikzpicture}[scale=0.8]
\footnotesize
\tikzstyle{every node}=[circle, inner sep=1.5pt];
\begin{scope}
\draw[thick] (0,0) node[draw] (p1) {$$};
\draw (-0.35,0) node {$p_1$};
\draw [line width=1pt, line cap=round, dash pattern=on 0pt off 3pt, rounded corners=5pt] (-0.8,-0.4) rectangle (0.5,0.4);
\draw[->] (1,0) -- (2.2,0);
\draw (1.6,0.3) node {0};
\draw (1.6,-0.3) node {\ap1};
\end{scope}
\begin{scope}[xshift = 100]
\draw[thick] (0,0) node[draw] (p1) {$$};
\draw[thick] (1,0) node[draw] (p2) {$$};
\draw (-0.35,0) node {$p_1$};
\draw (0.65,0) node {$p_2$};
\draw [line width=1pt, line cap=round, dash pattern=on 0pt off 3pt, rounded corners=5pt] (-0.8,-0.4) rectangle (1.5,0.4);
\draw[->] (2,0) -- (3.2,0);
\draw (2.6,0.3) node {1};
\draw (2.6,-0.3) node {\ap2};
\end{scope}
\begin{scope}[xshift = 230]
\draw[thick] (0,-0.6) node[draw] (p1) {$$};
\draw[thick] (1,-0.6) node[draw] (p2) {$$};
\draw[thick] (0.5,0.6) node[draw] (p3) {$$};
\draw (-0.35,-0.6) node {$p_1$};
\draw (0.65,-0.6) node {$p_2$};
\draw (0.15,0.6) node {$p_3$};
\draw (p1) -- (p3) -- (p2);
\draw [line width=1pt, line cap=round, dash pattern=on 0pt off 3pt, rounded corners=5pt] (-0.8,-1) rectangle (1.5,-0.2);
\draw [line width=1pt, line cap=round, dash pattern=on 0pt off 3pt, rounded corners=5pt] (-0.8,0.2) rectangle (1.5,1);
\draw[->] (2,0) -- (3.2,0);
\draw (2.6,0.3) node {2};
\draw (2.6,-0.3) node {\ap2};
\end{scope}
\begin{scope}[xshift = 355, yshift = -17]
\draw[thick] (0,-0.6) node[draw] (p1) {$$};
\draw[thick] (1,-0.6) node[draw] (p2) {$$};
\draw[thick] (0.5,0.6) node[draw] (p3) {$$};
\draw[thick] (0.5,1.8) node[draw] (p4) {$$};
\draw (-0.35,-0.6) node {$p_1$};
\draw (0.65,-0.6) node {$p_2$};
\draw (0.15,0.6) node {$p_3$};
\draw (0.15,1.8) node {$p_4$};
\draw (p1) -- (p3) -- (p2);
\draw (p3) -- (p4);
\draw [line width=1pt, line cap=round, dash pattern=on 0pt off 3pt, rounded corners=5pt] (-0.8,-1) rectangle (1.5,-0.2);
\draw [line width=1pt, line cap=round, dash pattern=on 0pt off 3pt, rounded corners=5pt] (-0.8,0.2) rectangle (1.5,1);
\draw [line width=1pt, line cap=round, dash pattern=on 0pt off 3pt, rounded corners=5pt] (-0.8,1.4) rectangle (1.5,2.2);
\end{scope}
\begin{scope}[xshift = 100, yshift = -150]
\draw[thick] (0,-0.6) node[draw] (p1) {$$};
\draw[thick] (1,-0.6) node[draw] (p2) {$$};
\draw[thick] (0.5,0.6) node[draw] (p3) {$$};
\draw[thick] (0.5,1.8) node[draw] (p4) {$$};
\draw[thick] (2,-0.6) node[draw] (p5) {$$};
\draw (-0.35,-0.6) node {$p_1$};
\draw (0.65,-0.6) node {$p_2$};
\draw (0.15,0.6) node {$p_3$};
\draw (0.15,1.8) node {$p_4$};
\draw (1.65,-0.6) node {$p_5$};
\draw (p1) -- (p3) -- (p2);
\draw (p3) -- (p4);
\draw [line width=1pt, line cap=round, dash pattern=on 0pt off 3pt, rounded corners=5pt] (-0.8,-1) rectangle (2.5,-0.2);
\draw [line width=1pt, line cap=round, dash pattern=on 0pt off 3pt, rounded corners=5pt] (-0.8,0.2) rectangle (2.5,1);
\draw [line width=1pt, line cap=round, dash pattern=on 0pt off 3pt, rounded corners=5pt] (-0.8,1.4) rectangle (2.5,2.2);
\draw[->] (-2.4,0.6) -- (-1.2,0.6);
\draw (-1.8,0.9) node {0};
\draw (-1.8,0.3) node {\ap1};
\end{scope}
\begin{scope}[xshift = 250, yshift = -167]
\draw[thick] (0,-0.6) node[draw] (p1) {$$};
\draw[thick] (1,-0.6) node[draw] (p2) {$$};
\draw[thick] (1,1.8) node[draw] (p3) {$$};
\draw[thick] (1,3.0) node[draw] (p4) {$$};
\draw[thick] (2,-0.6) node[draw] (p5) {$$};
\draw[thick] (0,0.6) node[draw] (p6) {$$};
\draw (-0.35,-0.6) node {$p_1$};
\draw (0.65,-0.6) node {$p_2$};
\draw (0.65,1.8) node {$p_3$};
\draw (0.65,3.0) node {$p_4$};
\draw (1.65,-0.6) node {$p_5$};
\draw (-0.35,0.6) node {$p_6$};
\draw (p1) -- (p3) -- (p2);
\draw (p5) -- (p3) -- (p4);
\draw (p1) -- (p6) -- (p2);
\draw [line width=1pt, line cap=round, dash pattern=on 0pt off 3pt, rounded corners=5pt] (-0.8,-1) rectangle (2.5,-0.2);
\draw [line width=1pt, line cap=round, dash pattern=on 0pt off 3pt, rounded corners=5pt] (-0.8,0.2) rectangle (2.5,1);
\draw [line width=1pt, line cap=round, dash pattern=on 0pt off 3pt, rounded corners=5pt] (-0.8,1.4) rectangle (2.5,2.2);
\draw [line width=1pt, line cap=round, dash pattern=on 0pt off 3pt, rounded corners=5pt] (-0.8,2.6) rectangle (2.5,3.4);
\draw[->] (-2.4,1.2) -- (-1.2,1.2);
\draw (-1.8,1.5) node {1};
\draw (-1.8,0.9) node {\ap3};
\end{scope}
\end{tikzpicture}
\end{center}
\caption{The recursive construction of \cref{exa:bap}.}
\label{fig:bapexample}
\end{figure}

\begin{figure}
\begin{center}
\begin{tikzpicture}[scale=0.5, rounded corners = 5pt]
\footnotesize
\begin{scope}[yshift=15, circle, xshift=15, inner sep=1.5pt];
\draw (-2.8,-1.0) rectangle (3.5,5);
\draw (-1.8,2.25) node {$P=$};
\draw[thick] (0,0) node[draw] (1) {$$};
\draw[thick] (1.5,0) node[draw] (2) {$$};
\draw[thick] (3,0) node[draw] (5) {$$};
\draw[thick] (1.5,3) node[draw] (3) {$$};
\draw[thick] (1.5,4.5) node[draw] (4) {$$};
\draw[thick] (0,1.5) node[draw] (6) {$$};
\draw (1) -- (6) -- (2) -- (3) -- (1);
\draw (4) -- (3) -- (5);
\draw (0,-0.5) node {\small$p_1$};
\draw (1.5,-0.5) node {\small$p_2$};
\draw (3,-0.5) node {\small$p_5$};
\draw (-0.6,1.5) node  {\small$p_6$};
\draw (0.9,3) node  {\small$p_3$};
\draw (0.9,4.5) node  {\small$p_4$};
\end{scope}
\begin{scope}[xshift = 75ex, yshift=15ex]
\draw (0,0) node {$\A = (0,0,1,2,0,1)$};
\draw (-3.2,-1) rectangle (3.2,1);
\end{scope}
\begin{scope}[yshift = -40ex, xshift=6ex]
\draw (-4.1,-2.2) rectangle (4,2.4);
\draw (0,0) node {$M = 
\begin{pmatrix} 2 & 1 & 0 & 0 \\ 0 & 0 & 0 & 1 \\ 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 1 \end{pmatrix}$};
\end{scope}
\begin{scope}[xshift = 75ex, yshift = -38ex]
\draw (0,0) node {$\pi = 521634$};
\draw (-3.2,-1) rectangle (3.2,1);
\end{scope}
\draw[->] (0.2,-1) -- (0.2,-3);
\draw (-0.5,-2) node {$\bpm$};
\draw[<-] (1.8,-1) -- (1.8,-3);
\draw (2.6,-2) node {$\bmp$};
\draw[->] (5.8,2.8) -- (7.8,2.8);
\draw (6.8,3.4) node {$\bpa$};
\draw[<-] (5.8,1.5) -- (7.8,1.5);
\draw (6.8, 0.9) node {$\bap$};
\draw[->] (12.2,-1) -- (12.2,-3);
\draw (13.0,-2) node {$\bas$};
\draw[->] (5.2,-2.8) -- (7.2,-0.8);
\draw (5.7, -1.3) node {$\bma$};
\draw[<-] (6.2,-3.7) -- (8.2,-1.7);
\draw (7.75, -3.05) node {$\bam$};
\end{tikzpicture}
\end{center}
\caption{An example of the bijections.}
\label{fig:examples}
\end{figure}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Restricted sets}\label{sec:restricted_sets}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

In this section we introduce and study a subset of ascent sequences that we term {\it{restricted ascent sequences}}.

\begin{definition}
Let $\RAsc_n$ be the subset of $\Asc$ consisting of those ascent sequences $\A=(\a_1,\ldots,\a_n)$ such that
\begin{equation}\label{equ:rasc_original}
\a_k \in [0,\a_{k-1}] \cup \{1+\asc(\a_1,\ldots,\a_{k-1})\} \mbox{\ \ for all $k>1$}.
\end{equation}
\end{definition}

In other words, if $\a_k$ is larger than $a_{k-1}$, then $\a_k$ must be the largest it can be under the conditions on an ascent sequence.  
For example, $(0,1,0,2)$ is in $\RAsc$ whereas $(0,1,0,1)$ is not.  

\cref{cor:restricted_bijections} is the main result of this section. It states that, under the classical bijections of ~\cite{bcdk,djk,dp}, $\RAsc_n$ maps to the sets we now define.

\begin{definition} \
\begin{itemize}
\item
Let $\RMatrices_n$ be the set of matrices in $\Matrices_n$ all of whose diagonal entries are positive.
\item 
Let $\RPosets_n$ be the set of those posets in $\Posets_n$ that have a chain of length $\level(P)$.  
\item 
Let $\RPerms_n  = \sn(3\overline{1}52\overline{4})$,  the set of length $n$ permutations $\pi$ such that every occurrence of the pattern 231 in $\pi$ plays the role of 352 in an occurrence of the pattern 31524 in $\pi$.
\end{itemize}
\end{definition}


The following observation is crucial for several results stated in this abstract.  
It not only gives the image of an element of $\RMatrices$ under $\bma$ but, combined with \cref{cor:restricted_bijections}(a), shows that all elements of $\RAsc$ take a particular form.  We use $i^j$ to denote a sequence of $j$ copies of $i$.

\begin{lemma}\label{rmattorasc}
Suppose that $M \in \RMatrices$ with $\dim(M)=d$.  Then 
\begin{equation}\label{equ:asc2matrix}
\bma(M) = (0^{m_{11}},1^{m_{22}},0^{m_{12}}, 2^{m_{33}},1^{m_{23}},0^{m_{13}},\ldots,(d-1)^{m_{dd}},(d-2)^{m_{d-1\,d}},\ldots,0^{m_{1d}}).
\end{equation}
\end{lemma}

\begin{example}
$M = \left(\begin{smallmatrix} 2&0&1 \\ 0&1&0 \\ 0&0&1\end{smallmatrix}\right)$ is in bijection with the ascent sequence $\A = (0,0,1,2,0)$.
\end{example}

The essential condition for the desired bijection between $\RAsc_n$ and $\RMatrices_n$ is as follows:  

\begin{prop}\label{pro:diag_zeros}
Let $\A=(\a_1,\ldots,\a_n) \in \Asc_n$ and $M=\bam(\A)$ with $\dim(M)=d$.
There will be a zero on the diagonal of $M$ if and only if there exists $i\in [1,n-1]$ such that $\a_i<\a_{i+1}\leq \asc(\a_1,\ldots,\a_i)$.
\end{prop}

We next provide a similarly essential ingredient for the desired bijection between $\RPosets$ and $\RMatrices$.

\begin{prop}\label{pro:rposets_to_rmatrices}
Let $P \in \Posets_n$ and let $M = \bpm(P)$ with $dim(M) = \level(P)+1$.  All the entries on the diagonal of $M$ will be non-zero if and only if $P$ has a chain of length $\level(P)$.
\end{prop}

Combining the previous two propositions gives the main result of this section, which shows that $\RAsc$, $\RMatrices$, $\RPosets$ and $\RPerms$ are all in bijection, and the bijections we need are exactly the restricted versions of the classical ones.  Part (c) of the theorem was already proved as~\cite[Prop.~10]{bcdk}

\begin{corollary}\label{cor:restricted_bijections}  The classical bijections among $\Asc_n$, $\Matrices_n$, $\Posets_n$ and $\Perms_n$ restrict to bijections among $\RAsc_n$, $\RMatrices_n$, $\RPosets_n$ and $\RPerms_n$, Specifically, 
\begin{enumerate}
\item $\bam$ maps $\RAsc_n$ bijectively to $\RMatrices_n$;
\item $\bpm$ maps $\RPosets_n$ bijectively to $\RMatrices_n$;
\item \cite{bcdk} $\bas$ maps $\RAsc_n$ bijectively to $\RPerms_n$.
\end{enumerate}
\end{corollary}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Poset duality under the bijections}\label{sec:duality}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

If a poset $P$ is (2+2)-free, then it is clear that the dual poset $P^*$ obtained by reversing all its inequalities is also (2+2)-free.  An open question in~\cite{dp} asks how $\A$ is related to $\A^*$, where $\A$ and $\A^*$ are the ascent sequences corresponding to $P$ and $P^*$ respectively.  While this question appears intractable for general (2+2)-free posets, in this section we answer it for $\RPosets$.  In addition, we extend the answer to give the corresponding notion of duality for $\RPerms$.  Combined with the duality result for $\RMatrices$ given by~\cite[Theorem~10]{djk}, we get a complete understanding of how poset duality acts on our four $\R$-families according to our bijections.  In fact, one major motivation for our restriction to the $\R$-families is their amenability to adopting an analogue of poset duality.  

In view of \cref{cor:restricted_bijections}, we can abuse notation by using the same $f$ notation for our bijections even though our domains will now be $\R$-families as opposed to the domains of $\Asc_n$, $\Matrices_n$, $\Posets_n$ and $\Perms_n$ that we had before. 

\begin{definition}
Let $f:\RPosets_n \to \Structure_n$ be a bijection where $\Structure_n$ is a collection of objects.
Given $P \in \RPosets$ with $f(P)=s$, we write $s^{*}$ for the unique object $f(P^{*})$, and we call $s^{*}$ the \emph{dual} of $s$ according to $f$.
\end{definition}

\begin{observation}\label{exa:matrix_dual}
As a first example, we consider the dual of an element $M$ of $\RMatrices$ according to the bijection $\bpm$.  Define $\flip{M}$ to be the reflection of $M$ through its antidiagonal, i.e., if $\dim(M)=d$, then $\flip{M}_{ij} = m_{d+1-j,\, d+1-i}$.  Observe that $M \in \RMatrices_n$ if and only if $\flip{M} \in \RMatrices_n$.  Theorem 10 from~\cite{djk} states that $M^* = \flip{M}$.
\end{observation}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsection{Duality for ascent sequences}\label{sub:dual_ascent_sequences}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

We will use \cref{exa:matrix_dual} as a basis for determining the dual of an element of $\RAsc$ according to $\bpa$.  

\begin{theorem}\label{thm:ascent_flip}
Let $\A \in \RAsc_n$.  By \cref{rmattorasc}, we have
\begin{equation}\label{equ:asc2matrix2}
\A = (0^{m_{11}},1^{m_{22}},0^{m_{12}}, 2^{m_{33}},1^{m_{23}},0^{m_{13}},\ldots,(d-1)^{m_{dd}},(d-2)^{m_{d-1\,d}},\ldots,0^{m_{1d}})
\end{equation}
where $M=\bam(\A)$. 
The dual ascent sequence $\A^*$ according to $\bpa$ is given by
\begin{multline}\label{equ:asc_dual}
\A^*= (0^{m_{dd}},1^{m_{d-1\, d-1}},0^{m_{d-1\, d}}, 2^{m_{d-2\, d-2}},1^{m_{d-2\,d-1}},0^{m_{d-2\, d}},\\
\ldots,(d-1)^{m_{11}},(d-2)^{m_{12}},\ldots,0^{m_{1d}}).
\end{multline}
\end{theorem}


\begin{example}
If $\A = (0,0,1,2,0)$, then $M = \left(\begin{smallmatrix} 2&0&1 \\ 0&1&0 \\ 0&0&1\end{smallmatrix}\right)$.  Thus $\flip{M} = \left(\begin{smallmatrix} 1&0&1 \\ 0&1&0 \\ 0&0&2\end{smallmatrix}\right)$, and so $\A^* = (0,1,2,2,0)$.
\end{example}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsection{Duality for permutations}\label{sub:perm_dual}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

We use the definition of the dual of an ascent sequence to determine the dual $\pi^*$ of an element $\pi$ of $\RPerms = \sn(3\overline{1}52\overline{4})$ according to $\bps \coloneqq \bas \circ \bpa$.

Our definition of $\pi^*$ requires the use of what are perhaps the three best known involutions on a permutation $\pi = (\pi_1, \ldots, \pi_n)$: its inverse $\pi^{-1}$, its \emph{reverse} $\reverse(\pi) \coloneqq (\pi_n, \ldots, \pi_1)$, and its \emph{complement} $\complement(\pi) \coloneqq (n+1-\pi_1, \ldots, n+1-\pi_n)$.  Together, these involutions allow us to state the result with the most technical proof of this abstract.

\begin{theorem}
Let $\pi \in \sn(3\overline{1}52\overline{4}) = \RPerms$. The dual permutation according to $\bps$ is given by $\pi^{*} =(\complement(\reverse(\pi)))^{-1}$.
\end{theorem}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{A Catalan restriction and series-parallel posets}\label{sec:catalan}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

In this section, we consider subsets of the $\R$-families whose cardinalities are given by the Catalan numbers; we thus use the $\C$ prefix for naming these subsets.  The study of these subsets is also motivated by the results of applying our bijections to these subsets, which allow us to draw connections between some natural families: pattern-avoiding ascent sequences, pattern-avoiding permutations, and series-parallel (2+2)-free posets.  Pattern avoidance in ascent sequences is studied from an enumerative perspective in \cite{ds}.

\begin{definition}
Let $\CAsc_n$ denote the set of $101$-avoiding ascent sequences of length $n$.
\end{definition}

As shown \cite{ds}, the cardinality of $\CAsc_n$ is the Catalan number $C_n$.  As an example, of the 15 ascent sequences of length 4, the only one containing $101$ is $0101$.  Thus $14 = C_4 = |\CAsc_4|$.
It is certainly not the case that $\CAsc_n = \RAsc_n$ since $01021 \in \RAsc_n \setminus \CAsc_n$.  However, we do have the following relationship.

\begin{prop}\label{pro:RcontainsS}
$\CAsc_n \subseteq \RAsc_n$.
\end{prop}

We next turn to determining the image of $\CAsc_n$ under our bijections.  We begin with posets since this case is perhaps the most interesting.  Recall from \cref{fig:summary} that $\CPosets_n$ denotes the series-parallel posets in $\Posets_n$.  In other words, $\CPosets_n$ consists of those posets with $n$ elements that are both $(2+2)$-free and N-free.

\begin{prop}\label{pro:casc}
$\bap: \CAsc_n \to \CPosets_n$, and $\bap$ is a bijection.
\end{prop}

We follow Jel\'inek \cite{JelFishburn} to define the family $\CMatrices_n$.

\begin{definition}
An SE-pair of a matrix $M \in \Matrices$ is a pair of non-zero entries $m_{ij}$ and $m_{i'j'}$ such that $i < i'$, $j<j'$ and $i' < j$.  We say that $M$ is \emph{SE-free} if it contains no SE-pair.
\end{definition}

In \cite[Lem.~1.2]{JelFishburn}, Jel\'inek shows that $P \in \Posets$ is series-parallel if and only if $\bpm(P)$ is SE-free.

\begin{definition}
Define $\CMatrices_n$ to be the subset of $\Matrices_n$ consisting of those elements that are SE-free.
\end{definition}

Finally, we determine the $\C$-family for permutations.  The answer is quite appealing, namely $\CPerms = \sn(231)$.

\begin{prop}
$\bas$ maps $\CAsc_n$ bijectively to $\CPerms$.
\end{prop}

The above results now give the following ``Catalan'' refinement of \cref{cor:restricted_bijections}

\begin{corollary}\label{cor:c} The classical bijections among $\Asc_n$, $\Matrices_n$, $\Posets_n$ and $\Perms$ restrict to bijections among $\CAsc_n$, $\CMatrices_n$, $\CPosets_n$ and $\CPerms$. Specifically, 
\begin{enumerate} 
\item $\bam$ maps $\CAsc_n$ bijectively to $\CMatrices_n$;
\item \cite{JelFishburn} $\bpm$ maps $\CPosets_n$ bijectively to $\CMatrices_n$;
\item $\bas$ maps $\CAsc_n$ bijectively to $\CPerms$.
\end{enumerate}
\end{corollary}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%

\printbibliography

\end{document}


